Research Paper



Prevalence and Factors Associated With Postural Abnormalities in Male Students of Tehran Universities: A Cross-sectional Study

Mohammed Said Shehada¹ (10), Noureddin Karimi^{1*} (10), Parastoo Baraghoosh¹ (10), Fariba Mohammadi² (10), Amena Ahmadi¹ (10)

1. Department of Physiotherapy, Faculty of Rehabilitation Sciences, University of Social Welfare and Rehabilitation Sciences, Tehran, Iran. 2. Department of Sport Injuries and Corrective Exercises, Sports Medicine Research Center, Sport Sciences Research Institute, Tehran, Iran.



Citation Shehada MS, Karimi N, Baraghoosh P, Mohammadi F, Ahmadi A. Prevalence and Factors Associated With Postural Abnormalities in Male Students of Tehran Universities: A Cross-sectional Study. Physical Treatments. 2023; 13(2):77-86. http://dx.doi.org/10.32598/ptj.13.2.560.1

doi): http://dx.doi.org/10.32598/ptj.13.2.560.1

Article info: Received: 12 Feb 2022 Accepted: 30 Mar 2022 Available Online: 01 Apr 2023

Keywords:

Prevalence, Postural abnormalities, Risk factors, Physical activity

ABSTRACT

Purpose: Postural abnormalities, such as forward head posture (FHP), posterior kyphosis, and lumbar lordosis, can lead to discomfort and health problems. This cross-sectional study was conducted to investigate the prevalence of these abnormalities and their association with age, body mass index (BMI), and physical activity among male students of Tehran universities.

Methods: A cross-sectional study was conducted on 400 students of Tehran University of Medical Science to evaluate the relationship between postural abnormalities and associated factors. Participants' demographic data and physical activity levels were collected using questionnaires, and spinal curvature was assessed using a spinal mouse device and photogrammetry. Statistical analysis was performed using IBM SPSS software, version 22, and a Pearson correlation coefficient was used to examine the relationship between variables.

Results: The study revealed a high prevalence of postural abnormalities, with FHP being the most prevalent (84.5%), followed by lumbar lordosis (79.8%), and posterior kyphosis (34.7%). Factors, such as age, BMI, waist-to-hip ratio, and physical activity were significantly associated with the occurrence of postural abnormalities.

Conclusion: A study on male university students showed that FHP, posterior kyphosis, and lordosis are common postural abnormalities with varying degrees of occurrence. FHP was the most prevalent type of abnormality. Age and BMI were correlated with the prevalence of postural abnormalities, and kyphosis was more common among younger students. BMI was also significantly related to the prevalence of postural abnormalities, and physical activity was associated with the prevalence of FHP.

* Corresponding Author:

Noureddin Karimi, Associate Professor.

Address: Department of Physiotherapy, Faculty of Rehabilitation Sciences, University of Social Welfare and Rehabilitation Sciences, Tehran, Iran. Phone: +98 (21) 22180039

.....

E-mail: karimi@uswr.ac.ir

Highlights

• Postural abnormalities are a prevalent issue among individuals, the incidence of which varies based on age, occupation, and lifestyle.

- It is essential to identify and address risk factors to prevent the development or progression of postural abnormalities.
- · Physical activity is essential to maintain good posture and prevent postural abnormalities.

Plain Language Summary

Poor posture can lead to health problems, such as back pain and musculoskeletal disorders. University students are especially at risk of developing poor posture due to prolonged sitting and studying. Our study focused on male students in Tehran City, Iran, to understand how common poor posture is and what factors are associated with it. By taking photographs and analyzing them, we identified different types of postural abnormalities, such as forward head posture (FHP), posterior kyphosis, and lumbar lordosis. The study showed that the prevalence of postural abnormalities among male students of Tehran University was high. The most common type of abnormality was forward head posture, followed by increased lumbar lordosis and posterior kyphosis. Overall, this study provides crucial insights into the prevalence and factors associated with postural abnormalities in male students of Tehran University. The findings have crucial implications for developing interventions to prevent or treat poor posture, ultimately contributing to better long-term health outcomes for university students.

1. Introduction

ostural abnormalities are a common problem among the general population, which if not addressed, can lead to various musculoskeletal problems [1]. The prevalence of these abnormalities is high in both developing and developed countries [2].

Among the most common postural abnormalities are forward head posture (FHP), kyphosis, and lumbar lordosis [3]. These abnormalities can cause various musculoskeletal problems, such as neck and back pain, headaches, and even breathing difficulties in severe cases [4].

Several factors have been identified as contributing to the development of postural abnormalities, including age, body mass index (BMI), and physical activity [5, 6]. In addition, evidence shows that certain occupations, such as those involving prolonged sitting, may increase the risk of postural abnormalities [7].

The prevalence of postural abnormalities in young adults, including university students, has been widely investigated. Some studies suggest that a high prevalence of postural deviations exists in young adults, with FHP being the most prevalent deviation [8-10]. In addition, previous studies have reported the relationship between spinal flexibility, kyphosis, and lordosis [11, 12]. Arsha-

di and Rajabi [11] found that increased spinal flexibility is associated with decreased thoracic kyphosis, while Samadi and Arshadi [12] reported that increased spinal flexibility is associated with decreased lumbar lordosis. Spinal curvature is also associated with body weight [13, 14] and physical activity level [15, 16]. Korovessis et al. [13] and Hoseinifar et al. [14] reported that individuals with greater body weight are more likely to have greater spinal curvature. On the other hand, Van Niekerk et al. [15] and Zagyapan et al. [16] found that individuals who engage in higher levels of physical activity are less likely to have spinal curvature.

Poor posture may lead to muscular imbalances [17]. Resistance exercises can improve postural deviations by strengthening the weak muscles that contribute to the deviation [18]. Therefore, a high level of physical activity may have a positive effect on postural abnormalities. In addition, several studies have reported a relationship between body weight and postural abnormalities. Rahimibashar and Motahari [19] reported a high prevalence of overweight and obesity among nursing students at the Islamic Azad University of Lahijan City, Iran. They suggested that being overweight and obese can contribute to the development of postural abnormalities.

Medical university students are a particularly crucial group to study because they are at high risk of musculoskeletal problems due to the demands of their training and long hours spent studying [20]. The current study used standardized measures to assess the prevalence of FHP, degree of kyphosis, and lumbar lordosis among male students in Tehran, City, Iran, medical universities. The study has also explored the potential relationship between these conditions and body weight and physical activity levels. By examining these factors, this study can provide valuable insights into the potential risk factors for spinal alignment issues in this population and inform future interventions aimed at improving spinal health and related outcomes.

Therefore, this study was conducted to investigate the prevalence of FHP, posterior kyphosis, and lumbar lordosis, and the relationship between these abnormalities with age, BMI, and physical activity in male students of Tehran universities.

2. Materials and Methods

This cross-sectional study was conducted on students in medical science universities (Tehran City, Iran, Shahid Beheshti, and Rehabilitation Sciences and Social Welfare University) during 2017-2018. Using Cochran's formula, 400 students were randomly selected as the research population by a two-stage clustering sampling method among 4233 students. In the first stage, clusters of students were selected based on their faculties. In the second stage, a random sample of 400 students aged 18-30 was selected from within each cluster. The selected students were invited to participate in the study, and those who agreed to participate were enrolled as participants. This approach helped to reduce the potential for bias in the selection of participants and improve the generalizability of the findings.

All participants in this study have read and signed the consent form. Participants were informed of the process and purpose of the study and had the right to withdraw from the study at any time if they were not satisfied.

Following the acquisition of written consent, participants' personal information, such as their name, age, and educational level, was documented in a questionnaire. Demographic information, including age, weight, BMI, and waist-to-hip ratio (WHR) were obtained, and subjects completed a physical activity questionnaire. The questionnaire used in the article includes two questions, have you engaged in any physical activities in the past week, such as walking, cycling, running, etc.?, how many minutes of physical activity have you had per session?, Physical activity less than 150 minutes per week is considered undesirable, 150-300 minutes is relatively desirable, and more than 300 minutes is desirable [21].

The objectives and method of the study were elucidated to the participants before the spinal curvature assessment. The study utilized the spinal mouse device, manufactured by Idiag AG, Voletswil Company in Switzerland, to measure the kyphosis and lordosis angles of the spine. The spinal mouse is a non-invasive device that has been approved by some researchers as highly reliable to measure spinal curvatures [22, 23]. The assessment was performed twice by the same examiner, who was experienced in using the spinal mouse. The equipment and assessment procedures were the same on both test days. Participants were instructed to stand comfortably with their head looking forward and focused on an eye-level marker, feet shoulder-width apart, knees straight, and hands hanging beside their body. The examiner marked the C7 spinous process and the top edge of the S2 vertebra, after which the spinal mouse device was placed on the defined C7 spinous process and moved along the spine's midline to the top of the second sacral vertebra at a constant speed. This process was repeated twice, and the mean was calculated and analyzed as the degree of kyphosis and lumbar lordosis for each subject.

To measure the angle of FHP, we employed photogrammetry [24], a method of measuring the position and orientation of objects using photographs or images. In this study, we captured a photograph of the participant's head and neck from a side angle, using a high-quality digital camera. The procedure involved identifying and marking two spinous process landmarks of the seventh cervical vertebra and the tragus of the ear, which was performed by a trained person. To ensure that the participant did not hold their head in a fixed position, we asked them to close their eyes and gently flex and extend their neck three times. This helped create a neutral position in the head area, which was crucial to eliminate any potential bias in the measurement. This process accounted for any natural variation in head position, ensuring accurate results. Finally, the angle of the FHP was calculated from the captured image while the participant stood in a natural position with their head looking straight ahead. The angle was measured as the distance between a line drawn from the seventh cervical vertebra to the tragus of the ear and a line drawn perpendicular to the ground. IBM SPSS software, version 22 was used for statistical tests. Central inclination and dispersion indices were used to describe the quantitative data. The Pearson correlation coefficient was used to examine the relationship between variables. All statistical tests were conducted at a 95% confidence level (P<0.05).

3. Results

The subjects' variables included the mean age, weight, BMI, waist circumference, physical activity, and the degree of FHP, posterior kyphosis, and lumbar lordosis, which are described in detail in Table 1. The information provided in Table 1 shows that the dispersion between the mean age, weight, BMI, physical activity, and the degree of forward head deviation is low, but the dispersion between the mean posterior kyphosis and lumbar lordosis is high.

Our study results showed that the highest frequency of abnormality was related to FHP. According to Table 2, 84.5%, 34.7%, and 79.8% of the samples had abnormalities of FHP, posterior kyphosis, and lumbar lordosis, respectively.

Based on Table 3, a significant correlation was observed between the age of the subjects and the occurrence of abnormalities in the kyphotic posture. (P=0.002). However, no significant difference was observed in the prevalence of FHP and lumbar lordosis between the two age groups (P>0.05).

Table 4 presents the distribution of subjects in 3 groups based on their BMI. A significant correlation was observed between the BMI of the subjects and the incidence of FHP, posterior kyphosis, and lumbar lordosis abnormalities ($P \le 0.05$).

In the current study, the WHR of the subjects was tested and recorded. Based on the collected data, a significant correlation was observed between the WHR of the subjects and the risk of certain health abnormalities, such as FHP. However, no significant difference was observed in the prevalence of posterior kyphosis and lumbar lordosis based on the WHR (Table 5).

The results showed a statistically significant relationship for FHP (P=0.04) between abnormality and physical activity level, with a higher proportion of individuals with abnormal FHP having a relatively desirable physical activity level compared to individuals with desirable physical activity levels. For posterior kyphosis and lumbar lordosis, no statistically significant relationship was observed between abnormality and physical activity level with P of 0.16 and 0.07, respectively (Table 6).

According to the results of this study, 84.5% of the students were affected by FHP. As can be seen in Table 7, 28.2% of the students under study were simultaneously affected by both FHP and posterior kyphosis. Additionally, 70.8% of the students were simultaneously affected by FHP and lumbar lordosis.

According to the results of this study, 34.7% of students were suffering from posterior kyphosis. As shown in Table 8, 28.2% of the students under study were simultaneously suffering from FHP and posterior kyphosis. Additionally, 24.5% of the students simultaneously had posterior kyphosis and lumbar lordosis.

According to the results of this study, 79.7% of students were diagnosed with lumbar lordosis. As can be seen in Table 9, 70.8% of the studied students were simultaneously affected by the abnormality of lumbar lordosis and FHP. Additionally, 24.5% of students were simultaneously affected by posterior pelvic tilt and lumbar lordosis.

Table 1. Descriptive	findings related	to quantitative	variables of men studied

Variables	Mean±SD
Age (y)	23.5±3.4
Weight (Kg)	73.95±1.16
BMI (Kg/m²)	23.75±3.56
WHR	0.86±0.09
Physical activity level	18.7±1.2
FHP	46.44±5.58
Posterior Kyphosis	42.72±16.07
Lumbar lordosis	15.35±10.1

Abbreviations: BMI: Body mass index; WHR: Waist-to-hip ratio; FHP: Forward head posture. PHYSICAL TREATMENTS

Abnormality		No (%)
Abhormanty	Normal	Abnormal
FHP	62(15.5)	338(84.5)
Posterior kyphosis	261(62.25)	139(34.7)
Lumbar lordosis	81(20.2)	319(79.8)
FHP: Forward head posture.		PHYSICAL TREATMENTS

Table 2. Frequency of postural abnormalities in the examined samples

FHP: Forward head posture.

Table 3. Prevalence of postural abnormalities by age group of participants

6 km a mm a life .		_			
Abnormality	<25		>	25	- P
	Normal	Abnormal	Normal	Abnormal	
FHP	52(13)	264(66)	10(2.5)	74(18.5)	0.31
Posterior kyphosis	194(48.5)	122(30.5)	67(16.7)	17(18.5)	0.002
Lumbar lordosis	68 (17)	248 (62)	13 (3.3)	71 (17.7)	0.22
HP: Forward head postur	e.			PHYSI	CAL TREATMENT

FHP: Forward head posture.

Table 4. Prevalence of postural abnormalities categorized by BMI of the participants

			No	. (%)			
	BMI						
Abnormality	≤1	18.5	18.	5-25	≥	25	Р
	Normal	Abnormal	Normal	Abnormal	Normal	Abnormal	
FHP	5(1.3)	8(2)	36(9)	227(56.8)	21(5.3)	103(25.8)	0.04
Posterior kyphosis	8(2)	5(1.3)	183(45.8)	80(20)	70(17.5)	54(13.5)	0.04
Lumbar lordosis	0	13(3.3)	37(9.3)	226(56.5)	44(11)	80(20)	≤0.001
/II: Body mass index; FH	I: Body mass index; FHP: Forward head posture. PHYSICAL TF						REATMENT

Table 5. Prevalence of postural abnormalities based on WHR

			No	. (%)			
	WHR						-
Abnormality	≤0	.94	0.95	i-0.99	2	2 1	Р
	Normal	Abnormal	Normal	Abnormal	Normal	Abnormal	
FHP	53(13.3)	299(74.9)	3(0.8)	35(8.8)	5(1.3)	4(1)	0.002
Posterior kyphosis	236(59.1)	116(29.1)	21(5.3)	17(4.3)	4(1)	5(1.3)	0.14
Lumbar lordosis	68(17)	284(71.2)	12(3)	26(6.5)	0	9(2.3)	0.06

WHR: Waist-to-hip ratio; FHP: Forward head posture.

PHYSICAL TREATMENTS

			No	. (%)			
Abnormality	Physical Activity Level						
Abnormanty	Unde	Undesirable Relatively Desirable		Des	Desirable		
	Normal	Abnormal	Normal	Abnormal	Normal	Abnormal	
FHP	42(10.5)	256(64)	20(5)	64(16)	7(1.7)	11(2.7)	0.04
Posterior kyphosis	115(28.7)	233(58.2)	26(6.5)	19(4.7)	4(1)	3(0.7)	0.16
Lumbar lordosis	65(16.2)	283(70.7)	15(3.75)	27(6.75)	2(0.5)	8(2)	0.07
FHP: Forward head pos	ture.					PHYSICAL TRE	

Table 6. Prevalence of postural abnormalities based on the level of physical activity of the subjects

Table 7. Prevalence of FHP and its relationship with other abnormalities

Aba anna litu	Posterior Kyphosis		Lumbar Lordosis		
Abnormality —	-	+	-	+	
FHP-, (%)	9	6.5	6.5	9	
FHP+, (%)	56.3	28.2	13.8	70.8	
Ρ	0.19		<0.001		

FHP: Forward head posture.

PHYSICAL TREATMENTS

Table 8. Prevalence of posterior kyphosis and its relationship with other abnormalities

Abnormality ———	F	FHP		Lumbar Lordosis		
	-	+	-	+		
Posterior kyphosis -, (%)	9	56.3	10	55.3		
Posterior kyphosis +, (%)	6.5	28.2	10.3	24.5		
Ρ	0.19		0.001			
		0.19	0.0	01		

FHP: Forward head posture.

PHYSICAL TREATMENTS

Table 9. Prevalence of lumbar lordosis and its association with other abnormalities

Abnormality		FHP		Posterior Kyphosis		
	-	+	-	+		
Lumbar lordosis -, (%)	6.5	13.8	10	10.3		
Lumbar lordosis +, (%)	9	70.8	55.3	24.5		
Ρ	(0.001	<0.	001		
EHP: Forward head posture			F	PHYSICAL TREATMENTS		

FHP: Forward head posture.

4. Discussion

In this study, the prevalence of FHP, posterior kyphosis, and lumbar lordosis among male students in Tehran universities was evaluated and compared with factors, such as age, weight, WHR, BMI, and physical activity. Our study results showed that the highest frequency of abnormality was related to FHP, and 84.5%, 79.8%, and 34.7% of the samples had an abnormality in FHP, lumbar lordosis, and posterior kyphosis, respectively.

The highest frequency of these abnormalities was observed in the age group under 25 years old. The lack of a significant correlation between age and the prevalence of FHP and lumbar lordosis in this study suggests that other factors may play a more significant role in the development of these conditions. Previous studies have identified several risk factors for FHP and lumbar lordosis, such as poor ergonomics, prolonged sitting, lack of physical activity, and obesity [4, 26].

In this study, the Mean±SD of BMI and WHR in students were 23.75±3.56 and 0.86±0.09, respectively, which were in the normal range. A significant correlation was also observed between the BMI of the subjects and the occurrence of some abnormal postures such as FHP, posterior kyphosis, and lumbar lordosis. Additionally, according to the collected data, a significant correlation was observed between the physical activity of the subjects and the risk of some abnormal postures, such as FHP. Furthermore, a significant correlation was observed between the WHR of the subjects and the occurrence of FHP, and these abnormalities were more common in students with a WHR below 0.94.

A study by Greve conducted among young adults found a correlation between BMI and postural balance [25]. Another study investigated the association between BMI and spinal deformities. It found that below-normal BMI is associated with the severity of spinal deformities, while above-normal BMI appears to have a protective effect. However, the specific relationship between BMI and kyphosis was not explicitly mentioned in this study [26]. A study by Rabieezadeh et al. showed that height, weight, and BMI could not be considered as an appropriate criterion to associate kyphosis and lordosis angles in students of Tehran [27].

Guo et al. demonstrated in their research that a BMI of more than 24 kg/m² or a WHR of more than 0.85 can lead to increased spinal curvature. They also observed a significant correlation between BMI and lumbar lordosis, but no correlation between height and weight and

lumbar lordosis when considered as independent variables [28]. These results are consistent with the findings of our study. In our study, a significant correlation was observed between the BMI of participants and the occurrence of postural abnormalities, such as FHP, kyphosis, and lumbar lordosis. Moreover, a significant correlation was observed between the prevalence of kyphosis and lumbar lordosis among the students.

A study by Abdullah et al. found a significant relationship between BMI and lordosis, and found no relationship between BMI and kyphosis [29]. The findings of the Malepe et al. study demonstrated no significant relationship between BMI and kyphosis as well as scoliosis [30].

Despite the low level of physical activity, the students' BMI was in the normal range, which may be due to reasons, such as the reception of health education in universities, cultural attitudes towards physical activity, and other environmental factors. Previous studies have also reported the relationship between physical activity and posture. A study by de Calcaterra et al. suggested that prevention programs should not only focus on reducing body weight but also on establishing correct postural habits from an early age of childhood and adulthood [31].

The results of our study showed that the highest prevalence of abnormalities among the research sample was FHP, followed by lumbar lordosis, and posterior kyphosis, respectively, at 84.5%, 79.8%, and 34.7% of the sample affected. In 2010 [32], the highest percentage of musculoskeletal disorders in Isfahan University students were reported in the following order, kyphosis at 34.42%, FHP at 94.38%, and lordosis at 56.29%. In the study by Malepe et al., kyphosis was observed the most common type of postural deviation among the University of Venda students [30].

Based on the results of the present study, the lowest prevalence of abnormality among male medical university students was kyphosis with a frequency of 34.7%. While, in the study conducted by Bahrami and Farhadi, they referred to the inequality of the shoulders as the lowest abnormality in the studied boys in Lorestan Province, Iran [33]. Therefore, it can be concluded that individuals may experience specific abnormalities and complications based on geographical, cultural, and so-cial differences.

Identifying these complications provides an opportunity for rehabilitation and physical education specialists to strive to correct them by providing specific movements and behavioral patterns.

The study's results also highlight the need for postural screening and education programs for students in medical universities. These programs should focus on the prevention and management of postural deviations and musculoskeletal pain, as well as the promotion of physical activity and healthy weight management.

5. Conclusion

The results of this study showed that in male university students participating in the present study, postural abnormalities, including FHP, posterior kyphosis, and lordosis, are significantly prevalent with varying degrees of occurrence. FHP was the most common type of postural abnormality observed among these students. Age and BMI were correlated with the prevalence of postural abnormalities among students, such that the prevalence of kyphosis was much higher among students younger than 25 years compared to students older than 25 years. A significant correlation was observed between the prevalence of postural abnormalities and the BMI of students. Additionally, a significant relationship was observed between the prevalence of FHP and the physical activity of students.

Limitations and future research

One limitation of this study is its cross-sectional design, which does not allow for causal inferences to be made. Additionally, the study only included male medical students in Tehran universities, which may limit the generalizability of the findings to other populations. Future research should employ longitudinal and comparative designs to investigate the causal relationships between postural alignment, body weight, physical activity, and musculoskeletal pain in diverse populations. Finally, future research should evaluate the effectiveness of postural screening and education programs in improving postural alignment and preventing musculoskeletal pain among other populations. In addition, we suggest further research to examine the long-term consequences of postural abnormalities.

Ethical Considerations

Compliance with ethical guidelines

The local Ethics Committee of the University of Social Welfare and Rehabilitation Sciences approved the whole research process (Code: 97/801/4017/).

Funding

This research was supported by the research project of the University of Social Welfare and Rehabilitation Sciences.

Authors' contributions

All authors contributed equally to preparing this article.

Conflict of interest

The authors declared no conflict of interest.

Acknowledgments

The authors would like to thank the students who participated in the study for their collaboration and medical universities in Tehran. The authors are grateful Physiotherapy Department of the University of Social Welfare and Rehabilitation Sciences.

References

- Woolf AD, Erwin J, March L. The need to address the burden of musculoskeletal conditions. Best Practice & Research. Clinical Rheumatology 2012; 26(2):183-224. [DOI:10.1016/j. berh.2012.03.005] [PMID]
- [2] Taleschian-Tabrizi N, Alinezhad F, Pezeshki MZ, Dastgiri S, Eftekharsadat B, Dolatkhah N. Prevalence of spinal deformities among school age children in Iran: A systematic review and meta-analysis. International Journal of Pediatrics. 2022; 10(7):16402-16. [DOI:10.22038/ijp.2022.65750.4960]
- [3] Griegel-Morris P, Larson K, Mueller-Klaus K, Oatis CA. Incidence of common postural abnormalities in the cervical, shoulder, and thoracic regions and their association with pain in two age groups of healthy subjects. Physical Therapy. 1992; 72(6):425-31. [DOI:10.1093/ptj/72.6.425] [PMID]
- [4] Quek J, Pua YH, Clark RA, Bryant AL. Effects of thoracic kyphosis and forward head posture on cervical range of motion in older adults. Manual Therapy. 2013; 18(1):65-71. [DOI:10.1016/j.math.2012.07.005] [PMID]
- [5] Skelton DA. Effects of physical activity on postural stability. Age and Ageing. 2001. 30(suppl_4):33-9. [DOI:10.1093/ageing/30.suppl_4.33] [PMID]

- [6] Swarnalatha, S., A. Sivashankari, and D. Malarvizhi, Correlation between body mass index and dynamic postural control among young healthy adults. Biomedical Research and Clinical Practice. 2018; 3(3):1-6. [DOI:10.15761/BRCP.1000171]
- [7] Daneshmandi H, Choobineh A, Ghaem H, Karimi M. Adverse effects of prolonged sitting behavior on the general health of office workers. Journal of Lifestyle Medicine. 2017; 7(2):69-75. [DOI:10.15280/jlm.2017.7.2.69] [PMID]
- [8] Brianezi L, Cajazeiro DC, Maifrino LB. Prevalence of postural deviations in school of education and professional practice of physical education. Journal of Morphological Sciences. 2017; 28(1):35-6. [Link]
- [9] Jeffries LJ, Milanese SF, Grimmer-Somers KA. Epidemiology of adolescent spinal pain: A systematic overview of the research literature. Spine. 2007; 32(23):2630-7. [DOI:10.1097/ BRS.0b013e318158d70b] [PMID]
- [10] Pasdar Y, Niazi P, Darbandi M, Khalvandi F, Izadi N. [Effect of physical activity on body composition and quality of life among women staff of Kermanshah University of Medical Sciences in 2013 (Persian)]. Journal of Rafsanjan University of Medical Sciences. 2015; 14(2):99-110. [Link]
- [11] Arshadi R, Rajabi R, Alizade MH. [Investigate the correlation between the spinal flexibility with degree of the kyphosis and lordosis (Persian)]. Research on Sports Science. 2007; 5(15):123-32. [Link]
- [12] Samadi H. [Investigating the relationship between the degree of kyphosis and psychological factors in male students of Tehran University (Persian)]. Journal of Fundamentals of Mental Health. 2008; 10(37):55-61. [Link]
- [13] Korovessis P, Koureas G, Papazisis Z. Correlation between backpack weight and way of carrying, sagittal and frontal spinal curvatures, athletic activity, and dorsal and low back pain in schoolchildren and adolescents. Journal of Spinal Disorders & Techniques, 2004; 17(1):33-40. [DOI:10.1097/00024720-200402000-00008] [PMID]
- [14] Hoseinifar M, Ghiasi F, Akbari A. The relationship between lumbar and thoracic curves with body mass index and low back pain in students of Zahedan University of Medical Sciences. Journal of Medical Sciences. 2007; 7(6):984-90. [DOI:10.3923/jms.2007.984.990]
- [15] van Niekerk SM, Louw Q, Vaughan C, Grimmer-Somers K, Schreve K. Photographic measurement of upper-body sitting posture of high school students: A reliability and validity study. BMC Musculoskeletal Disorders. 2008; 9:113. [DOI:10.1186/1471-2474-9-113] [PMID] [PMCID]
- [16] Zagyapan R, Iyem C, Kurkcuoglu A, Pelin C, Tekindal MA. The relationship between balance, muscles, and anthropomorphic features in young adults. Anatomy Research International. 2012; 2012:146063. [DOI:10.1155/2012/146063] [PMID] [PMCID]
- [17] Lewis JS, Green A, Wright C. Subacromial impingement syndrome: The role of posture and muscle imbalance. Journal of Shoulder and Elbow Surgery. 2005; 14(4):385-92. [DOI: 10.1016/j.jse.2004.08.007] [PMID]

- [18] Hrysomallis C, Goodman C. A review of resistance exercise and posture realignment. The Journal of Strength & Conditioning Research. 2001; 15(3):385-90. [DOI:10.1519/00124278-200108000-00022]
- [19] Rahimibashar M, Motahari M. [Assessment of overweight status, obesity and abdominal obesity among nursing students in Islamic Azad University of Lahijan (Persian)]. Iranian Journal of Nursing Research. 2013; 8(3):10-7. [Link]
- [20] Penkala S, El-Debal H, Coxon K. Work-related musculoskeletal problems related to laboratory training in university medical science students: A cross sectional survey. BMC Public Health. 2018; 18(1):1208. [DOI:10.1186/s12889-018-6125-y] [PMID]
- [21] Leavitt MO. 2008 Physical activity guidelines for Americans. WASHINGTON, D.C: US Department of Health and Human Services; 2008. [Link]
- [22] Livanelioglu A, Kaya F, Nabiyev V, Demirkiran G, Fırat T. The validity and reliability of "Spinal Mouse" assessment of spinal curvatures in the frontal plane in pediatric adolescent idiopathic thoraco-lumbar curves. European Spine Journal. 2016; 25(2):476-82. [DOI:10.1007/s00586-015-3945-7] [PMID]
- [23] Barrett E, McCreesh K, Lewis J. Reliability and validity of non-radiographic methods of thoracic kyphosis measurement: A systematic review. Manual Therapy. 2014; 19(1):10-7. [DOI:10.1016/j.math.2013.09.003] [PMID]
- [24] Salahzadeh Z, Maroufi N, Ahmadi A, Behtash H, Razmjoo A, Gohari M, et al. Assessment of forward head posture in females: Observational and photogrammetry methods. Journal of Back and Musculoskeletal Rehabilitation. 2014; 27(2):131-9. [DOI:10.3233/BMR-130426] [PMID]
- [25] Greve J, Alonso A, Bordini AC, Camanho GL. Correlation between body mass index and postural balance. Clinics (Sao Paulo). 2007; 62(6):717-20. [DOI:10.1590/s1807-59322007000600010] [PMID]
- [26] Hershkovich O, Friedlander A, Gordon B, Arzi H, Derazne E, Tzur D, Shamiss A, Afek A. Association between body mass index, body height, and the prevalence of spinal deformities. Spine Journal. 2014;14(8):1581-7. [DOI:10.1016/j. spinee.2013.09.034] [PMID]
- [27] Rabieezadeh A, Hovanloo F, Khaleghi M, Akbari A. The relationship of height, weight and body mass index with curvature of spine kyphosis and lordosis in 12-15-year old male adolescents of Tehran. Turkish Journal of Sport and Exercise. 2016; 18(3):42-6. [Link]
- [28] Guo JM, Zhang GQ, Alimujiang. [Effect of BMI and WHR on lumbar lordosis and sacrum slant angle in middle and elderly women (Chinese)]. Zhongguo Gu Shang=China Journal of Orthopaedics and Traumatology. 2008; 21(1):30-1. [PMID]
- [29] Abdullah AM, McDonald R, Jaberzadeh S. The effects of backpack load and placement on postural deviation in healthy students: A systematic review. International Journal of Engineering Research and Applications (IJERA). 2012; 2(6):466-81. [Link]
- [30] Malepe, M., et al., The relationship between postural deviations and body mass index among university students. Biomedical Research. 2015; 26(3):437-42. [Link]

- [31] Calcaterra V, Marin L, Vandoni M, Rossi V, Pirazzi A, Grazi R, et al. Childhood obesity and incorrect body posture: impact on physical activity and the therapeutic role of exercise. International Journal of Environmental Research and Public Health. 2022. 19(24):16728. [DOI: 10.3390/ijerph192416728] [PMID]
- [32] Kargarfard M, Mahdavi-Nejad R, Ghasemi GA, Rouzbehani R, Ghias M, Mahdavi-Jafari Z, et al. [Assessment of spinal curvature in Isfahan University Students (Persian)]. Journal of Isfahan Medical School. 2010; 27(102):762-76. [Link]
- [33] Bahrami M, Farhadi A. [A survey of rate and causes of deformity in boys and girls youngsters in Lorestan province (Persian)]. Yafte. 2007; 8(4):31-5. [Link]